% AMS-LaTeX 1.2 sample file for book proceedings, based on amsproc.cls.
%
\documentclass[twoside]{amsproc}
%\font\mynine=cmr9
\def\quocar{We follow \cite{carath}, pages 210-212.}
\newcounter{gnucount}
\newcommand{\Ft}[1]{\footnote{\mynine\baselineskip 8pt #1}}
\setcounter{gnucount}{0}
\newcommand{\gnu}[1]{\refstepcounter{gnucount}
\medbreak
\begin{center}
\fbox{\begin{minipage}{11cm}
\noindent GNU NO. \thegnucount.\ \ #1\ GNU-END
\end{minipage}}\end{center}
\medbreak}
%Uncomment the following  line to run file without printing the gnus
%\renewcommand{\gnu}[1]{}
\usepackage{epsf}
\input epsf
\usepackage{amssymb}
%\usepackage{makeidx}
\parskip=0.8mm
\def\comment#1{}
\def\snell{Snell}
\def\traj{{\rm Traj\,}}
\def\llist#1{
\vspace{-4.5mm}
\begin{center}
$\quad\quad\,\,$\mbox{\begin{minipage}{12.2cm}
#1
\end{minipage}}\end{center}}

\def\lllist#1{
\vspace{-4.5mm}
\begin{center}
$\quad\quad\quad$\mbox{\begin{minipage}{11.9cm}
#1
\end{minipage}}\end{center}}

\newcommand{\compo}{\circ}
\def\transv{{\cap\kern-5pt\lower -2.2pt\hbox{$|$}\kern-4.9pt\lower
-7pt\hbox{$-$}}} 
\def\baba{|\kern-1.7pt|}
\def\stransv{{\cap\kern-5.3pt\lower -2.2pt\hbox{$\baba$}\kern-5.3pt\lower
-7pt\hbox{$-$}}} 
\newcommand{\eps}{\varepsilon}
\def\verylongrightarrow{\raise2.2pt\hbox to 25pt{\hrulefill}\kern-4pt\longrightarrow}
\def\verylongline{\raise2.2pt\hbox to 37pt{\hrulefill}}
\def\ovarrow#1{{\stackrel{#1}{\verylongrightarrow}}}
\def\ovline#1{{\stackrel{#1}{\verylongline}}}
\def\ovlarrow#1{{\stackrel{#1}{\verylongleftarrow}}}
\def\verylongleftarrow{\longleftarrow\kern-4pt\raise2.2pt\hbox to 25pt{\hrulefill}}
\def\longsearrow{\setlength{\unitlength}{0.5cm}
\begin{picture}(1.2,1.2)
\put(-0.2,1.2){\vector(1,-1){2}}
\end{picture}}
\def\longnearrow{\setlength{\unitlength}{0.5cm}
\begin{picture}(1.2,1.2)
\put(-0.2,-1.2){\vector(1,1){2}}
\end{picture}}
\def\longdownarrow{\setlength{\unitlength}{0.25cm}
\begin{picture}(0.6,1)
\put(0.2,1){\vector(0,-1){2}}
\end{picture}}
\def\longuparrow{\setlength{\unitlength}{0.25cm}
\begin{picture}(1.2,1.2)
\put(0.2,-0.5){\vector(0,1){2}}
\end{picture}}
\def\veceps{\vec{\eps}} 
\def\vic{\pmb}
\def\Is{I^{\#}}
\def\Xis{\Xi^{\#}}
\def\xis{\xi^{\#}}
\def\etas{\eta^{\#}}
\def\zetas{\zeta^{\#}}
\def\qs{q^{\#}}
\def\ts{t^{\#}}
\def\taus{\tau^{\#}}
\def\nus{\nu^{\#}}
\def\restr{\,\lceil\,}
\def\corestr{\,\rceil\,}
\def\equaldef{{\buildrel \rm def \over =}}
\def\mybox{$\lower3.5pt\vbox{\hrule height2.6pt depth-2pt\hbox{\vrule width0.6pt\kern1pt\vbox{\kern1pt\phantom{nj}\kern1pt}\kern1.5pt\vrule width0.6pt}\hrule height2.6pt depth-2pt}$\medbreak}
\def\proofendbox{\enspace\hfill \mybox\bigskip}
\def\proofendsq{\hfill $\square$\medskip}
\def\proofend{\hfill$\diamondsuit\,\,$}
\def\pmb#1{\setbox0=\hbox{$#1$}%
  \kern-.025em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.025em\raise.0433em\box0 }
\def\pmbo#1{\setbox0=\hbox{$#1$}%
  \kern-.055em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.055em\raise.0433em\box0 }
%\def\bem{\bf\em }
\def\bem{\em }
\def\bemu{{\bem U}}
\def\tto{\longrightarrow\kern-9pt\longrightarrow}
\def\zz{\mathbb{Z}}
\def\qq{\mathbb{Q}}
\def\cc{\mathbb{C}}
\def\rr{\mathbb{R}}
\def\nn{\mathbb{N}}
\def\kk{\mathbb{K}}
\def\bmath{\mathbb{B}}
\def\smath{\mathbb{S}}
\def\zmath{\mathbb{Z}}
\newcommand{\nball}{{\bmath}^n}
\font\mynino=cmr9 scaled\magstephalf
\def\tub{\mathcal{T}}
\def\tubx{\mathcal{T}^X}
\def\btub{\bar\mathcal{T}}
\def\btubx{\bar\mathcal{T}^X}
\def\linx{{\rm Lin}(X,X)}
\def\linxxp{{\rm Lin}(X,X')}
\def\id{\kern.2em{\rm I}\kern-.56em{\rm 1}}
\def\idnpball{\id_{\partial\bmath_n}}
\def\so{{\rm So}}
\def\ta{{\rm Ta}}
\def\gr{{\rm Gr}}
\def\dom{{\rm Do}}
\def\im{{\rm Im}}
\def\togr{\,\,{\buildrel \rm igr \over \longrightarrow}\,\,}
\def\caF{\mathcal{F}}
%\def\spaco{\ \ \ \ \ \ }
\newcommand{\ssection}[1]{
%\ifodd \thepage%\clearpage
%\thispagestyle{empty}\hbox{$\,$}%\clearpage
%\thispagestyle{empty}\else%\clearpage
%\thispagestyle{empty}\fi
\hbox{$\,$}\vskip 0.3cm\section{#1}\vskip0.5cm}
\newcommand{\ssubsection}[1]{\subsection{#1}
\addcontentsline{toc}{section}{$\,\,\,\,$\arabic{section}.\arabic{subsection}
\ #1}}
\newcommand{\ben}{\begin{enumerate}}
\newcommand{\een}{\end{enumerate}}
\newtheorem{theorem}{Theorem}[subsection]
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{Fact}[theorem]{Fact}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{example}[theorem]{Example}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{xca}[theorem]{Exercise}

\numberwithin{equation}{section}

%    Absolute value notation
\newcommand{\abs}[1]{\lvert#1\rvert}

%    Blank box placeholder for figures (to avoid requiring any
%    particular graphics capabilities for printing this document).
\newcommand{\blankbox}[2]{%
  \parbox{\columnwidth}{\centering
%    Set fboxsep to 0 so that the actual size of the box will match the
%    given measurements more closely.
    \setlength{\fboxsep}{0pt}%
    \fbox{\raisebox{0pt}[#2]{\hspace{#1}}}%
  }%
}

\setcounter{tocdepth}{1}

\begin{document}
\begin{thebibliography}{A}

\bibitem{berkovitz}{Berkovitz,~L.~D., \textit{Optimal Control Theory.}
Springer-Verlag, New York, 1974.}

\bibitem{carath}{Carath\'eodory,~C., \textit{Calculus of Variations
and Partial Differential Equations of the First Order. Part II:
Calculus of Variations.} Eng. transl. by R.~B.~Dean, 2nd edition,
Holden Day, Inc., San Francisco, 1967. (Originally published as
\textit{Variationsrechnung und Partielle Differentialgleichungen
erster Ordnung.} B.~G.~Teubner, Berlin, 1935.)}

\bibitem{cesari}{Cesari, L., \textit{Optimization---Theory and Applications}.
Springer-Verlag, New York, 1983.}

\bibitem{hestenesbook}{Hestenes,~M.~R., \textit{Calculus of Variations
and Optimal Control Theory.} Wiley, 1966.}

%\bibitem{hestenes}{Hestenes,~M.~R., \textit{A general problem in the calculus
%of variations with applications to paths of least time.}  Rand Corporation
%RM-100, ASTIA Document No. AD-112382, 1950.}

\bibitem{leemarkus}{Lee,~E.~B., and L.~Markus, \textit{Foundations of
Optimal Control Theory.} Wiley, New York, 1967.}

\bibitem{pontrybook}{Pontryagin,~L.~S., V.~G.~Boltyanskii, R.~V.~Gamkrelidze,
and E.~F.~Mischenko, \textit{The Mathematical Theory of Optimal Processes.}
Wiley, New York, 1962.}

\end{thebibliography}



\newcommand{\hoo}{\raise 1.2pt\hbox{$\stackrel{o}{H}$}}
\newcounter{scholco}
\setcounter{scholco}{0}
\newcounter{apco}
\setcounter{apco}{0}
\newcounter{figurco}
\setcounter{figurco}{0}
\renewcommand{\thescholco}{\Roman{scholco}}
\newcommand{\scholium}[1]{\refstepcounter{scholco}
\subsection*{$\!\!\!\!\!\!\!\!$SCHOLIUM \thescholco.\ \ #1}
\addcontentsline{toc}{section}{$\qquad${Scholium \thescholco.\
    \ #1}}}
\newcommand{\figurecap}[1]{\refstepcounter{figurco}\bigskip
\begin{center}{\sc Figure \arabic{figurco}}.\ #1
\end{center}\bigskip}
\newcommand{\ssubsectionap}[1]{\refstepcounter{apco}
\subsection*{$\!\!\!\!\!\!\!\!$A.\arabic{apco}.\ \ #1}
\addcontentsline{toc}{section}{$\qquad$A.\arabic{apco}.\
    \ #1}}

%\section*{This is an unnumbered first-level section head}
%This is an example of an unnumbered first-level heading.

%\specialsection*{This is a Special Section Head}
%This is an example of a special section head%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\footnote{Here is an example of a footnote. Notice that this footnote
%text is running on so that it can stand as an example of how a footnote
%with separate paragraphs should be written.
%\par
%And here is the beginning of the second paragraph.}%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\gua{H.~J.~SUSSMANN}
\def\guo{THE MAXIMUM PRINCIPLE}

\pagestyle{myheadings}
\markboth{\gua}{\guo}


\def\restr{\lceil}
\def\pmb#1{\setbox0=\hbox{$#1$}%
  \kern-.025em\copy0\kern-\wd0
  \kern.05em\copy0\kern-\wd0
  \kern-.025em\raise.0144em\box0 
  \kern-.025em\raise.0288em\box0 
  \kern-.025em\raise.0433em\box0 }
%\renewcommand{\thesubsubsection}{\S$\,$\arabic{subsubsection}$\,$}
\def\equaldef{\,{\buildrel \rm def \over =}\,}
\def\hham{\mathcal{ H}}
\font\mytenrm=cmr10 scaled\magstephalf
\font\mytensl=cmsl10 scaled\magstephalf
\font\mytenit=cmti10 scaled\magstephalf
\font\forsection=cmbx10 scaled\magstep 0
\font\mynine=cmr9
\font\mynino=cmr9 scaled\magstephalf
\font\mit=cmti9
\font\myninebf=cmbx9
\font\mysmall=cmr7
\font\mysmallit=cmti7
\font\myverysmall=cmr5
\def\eps{\varepsilon}
\def\smalltyperm{\baselineskip=12.2pt \rm}
\def\smalltypesl{\baselineskip=12.2pt \sl}
\newcommand{\bi}{\begin{itemize}}
\newcommand{\ei}{\end{itemize}}
\newenvironment{Quoterm}{
\bigskip\begin{center}\begin{minipage}{11cm}\smalltyperm\noindent}
{\end{minipage}\end{center}\bigskip}
\newenvironment{Quotesl}{
\bigskip\begin{center}
\begin{minipage}{11cm}\smalltypesl\noindent}{\end{minipage}
\end{center}\bigskip}

\newcommand{\bqrm}{\begin{Quoterm}}
\newcommand{\eqrm}{\end{Quoterm}}
\newcommand{\bqsl}{\begin{Quotesl}}
\newcommand{\eqsl}{\end{Quotesl}}
\parindent 6mm
\newtheorem{ttheorem}{Theorem*}
\newtheorem{etheorem}{Theorem*}
\newtheorem{qtheorem}{Theorem*}
\newtheorem{myquestion}[qtheorem]{Open Problem}
\newenvironment{question}{\begin{myquestion}$\!\!\!$.\ \rm}{\end{myquestion}}
\newcommand{\bques}{\begin{question}}
\newcommand{\eques}{\end{question}}
\thispagestyle{empty}
\newcommand{\hdleft}{{\rm H. J. Sussmann}}
\newcommand{\hdright}{{\rm \hfil Hamilton and Weierstrass}}

%\setcounter{page}{-1}

\begin{center}
{\Huge{\bf THE PONTRYAGIN\\
$\,$\\
\vspace{-5mm}
MAXIMUM PRINCIPLE}}\\
$\,\,$\\
{\Large{\it H\'ector J. Sussmann} --- November 29, 2000}
\end{center}

\medbreak
\begin{center}
{\Large\bf CONTENTS}
\end{center}

\medbreak

\thispagestyle{empty}
\noindent 1.\ CARATH\'EODORY'S MISSED OPPORTUNITY\hfill 1 

\medbreak
\noindent 2.\ THE MAXIMUM PRINCIPLE\hfill 2

\noindent \ 2.1\ A classical version for fixed time 
interval problems \hfill 3

\noindent \ 2.2\ A classical version for variable 
time interval problems \hfill 5

\noindent \ 2.3\ An example \hfill 7

\medbreak


{\large 

\noindent The purpose of this note is to present a precise statement of a
technically simple version of the Pontryagin maximum principle. But
before we get there we will add one more example to our list of ``missed
opportunities,'' by showing how Carath\'eodory failed to discover the
Hamiltonian maximization condition because of his insistence on using the
``wrong'' Hamiltonian.


\ssection{Carath\'eodory's missed opportunity}\label{subsec-carat} 

\noindent {\bem Constantin Carath\'eodory} (1873-1950) published his
work on the calculus of variations in a 1935 book,
cf. \cite{carath}. His discussion of the excess function, and of how
Weierstrass' condition can be expressed in Hamiltonian form, is worth
quoting in detail\Ft{\quocar}, because it shows that the author
understood the importance of expressing the Weierstrass condition in
Hamiltonian form, but was unable to find its simple Hamiltonian
meaning because he was using an inappropriate Hamiltonian formalism.

Carath\'eodory uses $x_i$ for the position variables (that is, our $q^i$) 
and $y_i$ for the momenta (that is, our $p_i$). He works in the neighborhood
of a ``regular line \mbox{element}'' $e=(t,x_i,\dot x_i)$,   where the
``regularity'' assumption is the requirement that the Hessian matrix 
$\{L_{\dot x_i,\dot x_j}\}_{1\le i,j\le n}$ be positive definite
at $e$.   This condition guarantees that, near $e$, one can express
everything in ``canonical coordinates'' $(t,x_i,y_i)$, i.e., that
the map $(t,x,\dot x)\to (t,x,y)$ is invertible, if  we let
$y_i=\chi_i(t,x,\dot x)$, where 
$\chi_i(t,x,\dot x)=\frac{\partial L}{\partial \dot x_i}(t,x,\dot x)$.

He then expresses the Weierstrass condition, which involves 
the ``line element $(t,x_i,\dot x_i)$'' together with 
a nearby ``line element'' $(t,x_i,x'_i)$,  in terms of the corresponding 
canonical coordinates $(t,x_i,y_i)$ and 
$(t,x_i,y'_i)$.  He uses $H$ to represent what he calls
``the Hamiltonian'' (that is, our classical Hamiltonian), given by 
the formula $H=\sum_{i=1}^ny_i\dot x_i-L(t,x,\dot x)$, where 
$\dot x_i=\varphi_i(t,x,y)$, and $\varphi$ is the inverse 
function of $\chi$ (that is, $\varphi(t,x,\chi(t,x,u))=u$ 
and $\chi(t,x,\varphi(t,x,y))=y$). In addition, he uses 
$H'$ to represent the Hamiltonian evaluated at the other 
point $(t,x,y')$  in phase space, that is,
$H'=\sum_{i=1}^ny'_ix'_i-L(t,x,x')$, where 
$x'_i=\varphi_i(t,x,y')$, $y'_i=\chi_i(t,x,x')$.

With these notations, the Weierstrass excess function becomes 
\begin{eqnarray*}
\mathcal{E}(t,x,\dot x,x')&=&
L(t,x,x')-L(t,x,\dot x)-\frac{\partial L}{\partial\dot x}
(t,x,\dot x)\cdot(x'-\dot x)\\
&=&L(t,x,x')-L(t,x,\dot x)-y\cdot(x'-\dot x)\\
&=&y\cdot\dot x-L(t,x,\dot x)-\Big(y\cdot x'-L(t,x,x')\Big)\\
&=&y\cdot\dot x-L(t,x,\dot x)-\Big(y'\cdot x'-L(t,x,x')\Big)-(y-y')\cdot x'\\
&=&H-H'-(y-y')\cdot x'\,.
\end{eqnarray*} 
Since
\begin{eqnarray*}
\frac{\partial H'}{\partial y'_j}&=&
\frac{\partial\quad}{\partial y'_j}\Big(\sum_{i=1}^ny'_ix'_i-L(t,x,x')\Big)\\
&=&x'_j+\sum_{i=1}^ny'_i\frac{\partial\varphi_i}{\partial y'_j}(t,x,y')
-\frac{\partial\quad}{\partial y'_j}\ Big(L(t,x,\varphi(t,x,y'))\Big)\\
&=&x'_j+\sum_{i=1}^ny'_i\frac{\partial\varphi_i}{\partial y'_j}(t,x,y')
-\sum_{i=1}^n\frac{\partial L}{\partial x'_i}(t,x,x')
\frac{\partial\varphi_i}{\partial y'_j}(t,x,x')\\
&=&x'_j+\sum_{i=1}^ny'_i\frac{\partial\varphi_i}{\partial y'_j}(t,x,y')
-\sum_{i=1}^ny'_i\frac{\varphi_i}{\partial y'_j}(t,x,x')\\
&=&x'_j\,,
\end{eqnarray*} 
Carath\'eodory concluded that 
``the equation
\begin{eqnarray*}
\mathcal{E}(t,x,\dot x,x')&=&L'-L-L_{\dot x_j}(x'_j-\dot x_j)\\
&=&H-H'-H'_{y'_j}(y_j-y'_j)
\end{eqnarray*} 
therefore holds, by which the $\mathcal{E}$ function is represented
in the desired form.''


\ssection{The maximum principle}

\noindent  The first rigorous statement and proof of the maximum principle
appears in the book \cite{pontrybook}. This ``classical'' version was
then improved by other authors, e.g. L. Cesari \cite{cesari},
M. Hestenes \cite{hestenesbook}, E. B. Lee and L. Markus
\cite{leemarkus}, and L. D. Berkovitz \cite{berkovitz}.  

Here we shall present a couple of simple versions of the result, 
under technical assumptions that will be considerably weakened later.

\ssubsection{A classical version for fixed time 
interval problems}

\noindent We begin by quoting a relatively simple version.

We assume that the following conditions are satisfied:

\medbreak\begin{itemize}
\item[C1.] $n$, $m$ are nonnegative integers;

\item[C2.] $Q$ is an open subset of $\rr^n$;

\item[C3.] $U$ is a closed subset of $\rr^m$;

\item[C4.] $a$, $b$ are real numbers such that $a\le b$;

\item[C5.] $Q\times U\times[a,b]\ni(x,u,t)\mapsto f(x,u,t)=
(f^1(x,u,t),\ldots,f^n(x,u,t)\in\rr^n$ and 
$Q\times U\times[a,b]\ni(x,u,t)\mapsto L(x,u,t)\in\rr$ are continuous maps; 

\item[C6.] for each $(u,t)\in U\times[a,b]$ the maps 
$$
Q\ni x\mapsto f(x,u,t)=\Big(f^1(x,u,t),\ldots,f^n(x,u,t)\Big)\in\rr^n
$$
and 
$$
Q\ni x\mapsto L(x,u,t)\in\rr
$$
are continuously differentiable, 
and their partial derivatives with respect to the $x$ coordinates
are continuous functions of $(x,u,t)$;

\item[C7.] $\bar x$, $\hat x$ are given points of $Q$;

\item[C8.] $TCP_{[a,b]}(Q,U,f)$ (the set of all ``trajectory-control pairs
defined on $[a,b]$ for the data $Q,U,f$'') is the set of all pairs
$(\xi,\eta)$ such that:

\begin{itemize}
\item[a.] $[a,b]\ni t\mapsto\eta(t)\in U$ is a measurable bounded map,

\item[b.] $[a,b]\ni t\mapsto\xi(t)\in Q$ is an absolutely continuous map,

\item[c.] $\dot\xi(t)=f(\xi(t),\eta(t),t)$ for almost every $t\in[a,b]$;.
\end{itemize}

\item[C9.] $TCP_{[a,b]}(Q,U,f)\ni(\xi,\eta)\mapsto J(\xi,\eta)\in\rr$ is the 
functional given by 
$$
J(\xi,\eta)\equaldef\int_a^bL(\xi(t),\eta(t),t)\,dt\,;
$$

\item[C10.] $\gamma_*=(\xi_*,\eta_*)$ (the ``reference TCP'') is
such that 

\begin{itemize}
\item[a.] $\gamma_*\in TCP_{[a,b]}(Q,U,f)$,

\item[b.] $\xi_*(a)=\bar x$ and $\xi_*(b)=\hat x$,

\item[c.] $J(\xi_*,\eta_*)\le J(\xi,\eta)$ for all 
$(\xi,\eta)\in TCP_{[a,b]}(Q,U,f)$ such that 
$\xi(a)=\bar x$ and $\xi(b)=\hat x$.
\end{itemize}
\end{itemize}

\begin{theorem}\label{theo-timenovar}
 Assume that the data $n$, $m$, $Q$, $U$, $a$, $b$, $f$, 
$L$, $\bar x$, $\hat x$ satisfy conditions C1-C7, $TCP_{[a,b]}(Q,U,f)$ 
and $J$ are defined by C8-C9, and $\gamma_*=(\xi_*,\eta_*)$ satisfies C10.
Define the Hamiltonian $H$ to be the function
$$
Q\times U\times\rr^n\times\rr\times[a,b]\ni(x,u,p,p_0,t)
\mapsto H(x,u,p,p_0,t)\in\rr
$$ 
given by
$$
H(x,u,p,p_0,t)\equaldef\langle p,f(x,u,t)\rangle-p_0 L(x,u,t)\,.
$$
Then there exists a pair $(\pi,\pi_0)$ such that
\begin{itemize}
\item[E1.] $[a,b]\ni t\mapsto\pi(t)\in\rr^n$ is an absolutely continuous
map;
\item[E2.] $\pi_0\in\rr$ and $\pi_0\ge 0$;
\item[E3.] $(\pi(t),\pi_0)\neq(0,0)$ for every $t\in[a,b]$;
\item[E4.] the ``adjoint equation'' holds, that is, 
$$
\dot\pi(t)=-\frac{\partial H}{\partial x}(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)
$$
for almost every $t\in[a,b]$;
\item[E5.] the ``Hamiltonian maximization condition'' holds, that is,
$$
H(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)=
\max\Big\{H(\xi_*(t),u,\pi(t),\pi_0,t):u\in U\Big\}
$$
for almost every $t\in[a,b]$.
\end{itemize} 
\end{theorem}

\begin{remark} {\rm
It is clear that
$$
\frac{\partial H}{\partial p}(x,u,p,p_0,t)=f(x,u,t)\,.
$$
Therefore the adjoint equation, together with the equation of C8-c, say that
the pair $(\xi_*,\pi)$ is a solution of  {\bem Hamilton's equations}}

\noindent\begin{minipage}{9.5cm}
\begin{eqnarray}
\dot\xi_*(t)&=&\phantom{-}\frac{\partial H}{\partial p}
(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)\,,\\
\dot\pi(t)&=&-\frac{\partial H}{\partial x}(\xi_*(t),\eta_*(t),\pi(t),
\pi_0,t)\,.
\end{eqnarray}\end{minipage}\begin{minipage}{2.8cm}
\null$\quad$\\
\null$\quad$\\
\null$\qquad\quad\,\,\qquad\quad\diamondsuit$\end{minipage}
\end{remark}
\begin{remark}{\rm 
The map $\pi$ is often referred to in the control theory literature as 
the ``adjoint vector.'' Its proper status from the differential-geometric 
point of view is that of a {\bem field of covectors} along $\xi_*$.} \proofend
\end{remark}
\begin{remark}{\rm 
The number $\pi_0$ is the ``abnormal multiplier.''}\proofend
\end{remark}
\begin{remark}{\rm 
A trajectory-control pair for which there exists a pair $(\pi,\pi_0)$ 
that satisfies E1-E5 is 
called an {\bem extremal}. Therefore the Maximum Principle 
says that a necessary condition for a TCP to be a solution of the problem of 
minimizing the functional $J(\xi,\eta)$ subject to the conditions 
$(\xi,\eta)\in TCP_{[a,b]}(Q,U,f)$ and the endpoint conditions
C10-b is that $(\xi,\eta)$ be an extremal.}\proofend
\end{remark}
\begin{remark}{\rm 
An extremal is {\bem normal} if the pair $(\pi,\pi_0)$ can be chosen so that
$\pi_0\neq 0$, and {\bem abnormal} if the pair $(\pi,\pi_0)$ can be chosen 
so that $\pi_0=0$, The pair $(\pi,\pi_0)$ need not be unique, so in particular
it can happen that an extremal is both normal and abnormal.}\proofend
\end{remark}
\begin{remark}{\rm 
Property E3 is referred to as the ``nontriviality condition.''}\proofend
\end{remark}
\begin{remark}{\rm 
The adjoint equation says that
$$
\dot\pi(t)=-\pi(t)\cdot
\frac{\partial f}{\partial x}(\xi_*(t),\eta_*(t),t)+\pi_0
\frac{\partial L}{\partial x}(\xi_*(t),\eta_*(t),t)\,.
$$
In particular, if $(\pi(t),\pi_0)=(0,0)$ for one value $\tau$ of $t$, 
then $\pi_0=0$, so the equation reduces to 
$$
\dot\pi(t)=-\pi(t)\cdot
\frac{\partial f}{\partial x}(\xi_*(t),\eta_*(t),t)\,,
$$
which is a linear homogeneous time-varying ordinary differential equation for 
$\pi$. Therefore the fact that $\pi(\tau) =0$ implies that $\pi(t)=0$ for 
all $t$, so $(\pi(t),\pi_0)=(0,0)$ for all $t$. Hence the nontriviality 
condition could equally well have been stated by saying ``for some $t$'' 
rather than ``for all $t$.''}\proofend
\end{remark}
\begin{remark}\label{pointed}{\rm 
If we let 
$$
h(t)=H(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)\,,
$$
then, formally, the derivative of $h$ is given by
\begin{eqnarray*}
\dot h(t)&=&\frac{\partial H}{\partial x}(\Gamma(t))\cdot\dot\xi_*(t)\!+\!
\frac{\partial H}{\partial u}(\Gamma(t))\cdot
\dot\eta_*(t)\!+\!\frac{\partial H}{\partial p}(\Gamma(t))\cdot\dot\pi(t)
\!+\!\frac{\partial H}{\partial t}(\Gamma(t))\\
&=&\frac{\partial H}{\partial u}(\Gamma(t))\cdot\dot\eta_*(t)
+\frac{\partial H}{\partial t}(\Gamma(t))\,,
\end{eqnarray*}
where we have written $\Gamma(t)=(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)$
and used the fact that $\dot\xi_*=\frac{\partial H}{\partial p}$ 
and $\dot\pi=-\frac{\partial H}{\partial x}$ by Hamilton's equations. 
If we disregard the fact that $\dot\eta_*(t)$ need not exist
(because $\eta_*$ is not assumed to be diferentiable or even continuous),
and in addition we pretend that 
``$\frac{\partial H}{\partial u}(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)$
must vanish because the function $u\mapsto H(\xi_*(t),u,\pi(t),\pi_0,t)$
has a maximum at $u=\eta_*(t)$,'' then we end up with 
$$
\dot h(t)=\frac{\partial H}{\partial t}(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)\,.
$$
This leads to the conjecture---that we are very far from having proved---that
{\bem if the dynamical law $f$ and the Lagrangian $L$ do not depend on 
$t$ then the function
$[a,b]\ni t\mapsto H(\xi_*(t),\eta_*(t),\pi(t),\pi_0,t)$ is constant.}
We will see later that this can be made precise and proved rigorously.}
\end{remark}

\ssubsection{A classical version for variable time 
interval problems}

\noindent We assume that the following conditions are satisfied:

\medbreak\begin{itemize}
\item[C1.] $n$, $m$ are nonnegative integers;

\item[C2.] $Q$ is an open subset of $\rr^n$;

\item[C3.] $U$ is a closed subset of $\rr^m$;

\item[C4.] the maps $Q\times U\ni(x,u)\mapsto f(x,u)=
(f^1(x,u),\ldots,f^n(x,u))\in\rr^n$ and 
$Q\times U\ni(x,u)\mapsto L(x,u)\in\rr$ are continuous; 

\item[C5.] for each $u\in U$ the maps 
$$
Q\ni x\mapsto f(x,u)=\Big(f^1(x,u),\ldots,f^n(x,u)\Big)\in\rr^n
$$
$$
\!{\rm and}\qquad\qquad\qquad Q\ni x\mapsto L(x,u)\in\rr\qquad
\qquad\qquad\qquad\qquad
$$
are continuously differentiable, 
and their partial derivatives with respect to the $x$ coordinates
are continuous functions of $(x,u)$;

\item[C6.] $\bar x$, $\hat x$ are given points of $Q$;

\item[C7.] $TCP(Q,U,f)$ (the set of all ``trajectory-control pairs
for the data $Q,U,f$'') is the set given by 
$$
TCP(Q,U,f)\equaldef\bigcup
\Big\{TCP_{[a,b]}(Q,U,f):a,b\in\rr,a\le b\Big\}\,,
$$
where, for $a,b\in\rr$ such that $a\le b$, 
$TCP_{[a,b]}(Q,U,f)$ is the set of all pairs
$(\xi,\eta)$ such that:

\begin{itemize}
\item[a.] $[a,b]\ni t\mapsto\eta(t)\in U$ is a measurable bounded map;

\item[b.] $[a,b]\ni t\mapsto\xi(t)\in Q$ is an absolutely continuous map;

\item[c.] $\dot\xi(t)=f(\xi(t),\eta(t))$ for almost every $t\in[a,b]$.
\end{itemize}

\item[C8.] $TCP(Q,U,f)\ni(\xi,\eta)\mapsto J(\xi,\eta)\in\rr$ is the 
functional given by 
$$
J(\xi,\eta)\!\equaldef\!\int_a^bL(\xi(t),\eta(t))\,dt\,\,\,{\rm for}\,\,
(\xi,\eta)\!\in\! TCP_{[a,b]}(Q,U,f),\,a,b\!\in\!\rr,\,a\le b.
$$

\item[C9.] $\gamma_*=(\xi_*,\eta_*)$ (the ``reference TCP'') are
$a_*$, $b_*$ are such that 

\begin{itemize}
\item[a.] $a_*\in\rr$, $b_*\in\rr$, and $\gamma_*\in TCP_{[a_*,b_*]}(Q,U,f)$;

\item[b.] $\xi_*(a_*)=\bar x$ and $\xi_*(b_*)=\hat x$;

\item[c.] $J(\xi_*,\eta_*)\le J(\xi,\eta)$ for all $a,b\in\rr$ such that
$a\le b$ and all $(\xi,\eta)\in TCP_{[a,b]}(Q,U,f)$ such that 
$\xi(a)=\bar x$ and $\xi(b)=\hat x$.
\end{itemize}
\end{itemize}

\begin{theorem}\label{theo-timevar}
Assume that the data $n$, $m$, $Q$, $U$, $f$, 
$L$, $\bar x$, $\hat x$ satisfy conditions C1-C6, $TCP(Q,U,f)$ 
and $J$ are defined by C7-C8, and $a_*$, $b_*$,  
$\gamma_*=(\xi_*,\eta_*)$ satisfy C9.
Define the Hamiltonian $H$ to be the function
$$
Q\times U\times\rr^n\times\rr\ni(x,u,p,p_0)
\mapsto H(x,u,p,p_0)\in\rr
$$ 
given by
$$
H(x,u,p,p_0)\equaldef\langle p,f(x,u)\rangle-p_0 L(x,u)\,.
$$
Then there exists a pair $(\pi,\pi_0)$ such that
\begin{itemize}
\item[E1.] $[a_*,b_*]\ni t\mapsto\pi(t)\in\rr^n$ is an absolutely continuous
map;
\item[E2.] $\pi_0\in\rr$ and $\pi_0\ge 0$;
\item[E3.] $(\pi(t),\pi_0)\neq(0,0)$ for every $t\in[a_*,b_*]$;
\item[E4.] the ``adjoint equation'' holds, that is, 
$$
\dot\pi(t)=-\frac{\partial H}{\partial x}(\xi_*(t),\eta_*(t),\pi(t),\pi_0)
$$
for almost every $t\in[a_*,b_*]$;
\item[E5.] the ``Hamiltonian maximization condition with zero value'' 
holds, that is,
$$
0=H(\xi_*(t),\eta_*(t),\pi(t),\pi_0)=
\max\Big\{H(\xi_*(t),u,\pi(t),\pi_0):u\in U\Big\}
$$
for almost every $t\in[a,b]$.
\end{itemize} 
\end{theorem}

\begin{remark}{\rm
If we fix $a_*$ and $b_*$, and consider only TCPs $(\xi,\eta)$ belonging
to $TCP_{[a_*,b_*]}(Q,U,f)$, then $\gamma_*$ is obviously a solution
of the minimization problem considered in Theorem 
\ref{theo-timenovar}. So all the conditions given by that theorem
should be satisfied here. And, indeed, this is what happens, because
{\bem the only difference between the conclusion 
of Theorem \ref{theo-timenovar} and that 
of Theorem \ref{theo-timevar} is that in 
Theorem \ref{theo-timevar} we obtain the additional condition that the
value of the Hamiltonian must vanish.}}\proofend
\end{remark}
\begin{remark}{\rm
We already pointed out in Remark \ref{pointed} that, at least formally, 
the conclusions of Theorem \ref{theo-timenovar} already imply 
that the Hamiltonian function
$t\mapsto H(\xi_*(t),\eta_*(t),\pi(t),\pi_0)$ is constant, since in
our problem $f$ and $L$ do not depend on $t$. So the only novelty 
of Theorem \ref{theo-timevar} is that the constant now turns out to be equal 
to zero.}\proofend
\end{remark}

\ssubsection{An example}

We consider the ``one-dimensional soft landing'' problem in which it is 
desired to find, for a given initial point $(\alpha,\beta)\in\rr^2$,
a trajectory-control pair $(\xi,\eta)$ of the system 
$$
\dot x=y\,,\qquad\dot y=u\,,\qquad -1\le u\le 1\,,
$$
such that $\xi$ goes from $(\alpha,\beta)$ to $(0,0)$ in minimum time.

This is, of course, a problem of the kind discussed in Theorem 
\ref{theo-timevar}, so we shall solve it by applying the theorem.

In this case, the configuration space $Q$ is $\rr^2$, the control set
$U$ is the compact interval $[-1,1]$, the dynamical law $f$ is given
by 
$$
f(x,y,u)=(y,u)\,,
$$
and the Lagrangian is identically equal to $1$. 

The Hamiltonian $H$ is given by 
$$
H(x,y,u,p_x,p_y,p_0)=p_xy+p_yu-p_0\,,
$$
where we are using $p_x$, $p_y$ to denote the two components
of the momentum variable $p$. 

Assume that $(\xi_*,\eta_*)$ is a solution of our minimum time problem,
and that $(\xi_*,\eta_*)\in TCP_{[a_*,b_*]}(Q,U,f)$.

Then Theorem \ref{theo-timevar} tells us that there exists a pair 
$(\pi,\pi_0)$ satisfying all the conditions of the conclusion.
Write $\pi(t)=(\pi_x(t),\pi_y(t))$. The adjoint equation
then implies 
\begin{eqnarray*}
\dot\pi_x(t)&=&0\,,\\
\dot\pi_y(t)&=&-\pi_x(t)\,.
\end{eqnarray*}
Therefore the function $\pi_x$ is constant. (Notice that we are using the fact 
that $\pi_x$ is absolutely continuous!) Let $A\in\rr$ be such that
$\pi_x(t)=A$ for all $t\in[a_*,b_*]$. Then there must exist a constant
$B$ such that
$$
\pi_y(t)=B-At\quad{\rm for}\quad t\in[a_*,b_*]\,.
$$
Now, if $A$ and $B$ were both equal to zero, the function 
$\pi_y$ would vanish identically and then the Hamiltonian 
maximization condition would say that the function
$$
[-1,1]\ni u\mapsto 0
$$
is maximized by taking $u=\eta_*(t)$, a fact that would give us no 
information whatsoever about $\eta_*$. 

Fortunately, the conditions of Theorem \ref{theo-timevar} 
imply that {\bem $A$ and $B$ cannot both vanish.} To see this, observe 
that if $A=B=0$ then it follows that $\pi_x(t)\equiv\pi_y(t)\equiv 0$. 
But then the nontriviality condition tells us that $\pi_0\neq 0$. 
So the value $H(\xi_*(t),\eta_*(t),\pi(t),\pi_0)$ would be equal to 
$-\pi_0$, which is not equal to zero. This contradicts the fact that,
for our time-varying problem, the Hamiltonian is supposed to vanish.

Now that we know that $A$ and $B$ cannot both vanish, there are two
possibilities. Either $A=0$ or $A\neq 0$. 

Suppose first that $A=0$. Then $B\neq 0$, and
Hamiltonian maximization tells us that $[-1,1]\ni u\mapsto Bu$ is
maximized by $u=\eta_*(t)$. If $B>0$, this implies that $\eta_*(t)=1$
for all $t$. If $B<0$ then it follows that $\eta_*(t)=-1$ for all $t$.

Now assume that $A\neq 0$. Then $t\mapsto B-At\equaldef\varphi(t)$ is
a nonconstant linear function of $t$. Therefore $\varphi$ vanishes at
most once on the interval $[a_*,b_*]$. If $\varphi$ never vanishes on
$[a_*,b_*]$, or vanishes at one of the endpoints, then $\eta_*(t)$ is
either always equal to $1$ (if $\varphi(t)>0$) or always equal to $-1$
(if $\varphi(t)<0$).  If $\varphi(\tau)=0$ for some
$\tau\in\,]a_*,b_*[$, then $\eta_*(t)$ will be equal to $1$ for
$t<\tau$ and to $-1$ for $t>\tau$ (if $\varphi$ changes sign at $\tau$
from positive to negative) or 
$\eta_*(t)$ will be equal to $-1$ for
$t<\tau$ and to $1$ for $t>\tau$ (if $\varphi$ changes sign at $\tau$
from negative to positive).  

So we have proved that $\eta_*$ is of one of the following 
four types:

\medbreak
\begin{itemize}
\item[a.] constantly equal to $1$,

\item[b.] constantly equal to $-1$,

\item[c.] constantly equal to $-1$ for $t<\tau$ and to $1$ for 
$t>\tau$ for some $\tau\in\,]a_*,b_*[$, 

\item[d.] constantly equal to $1$ for $t<\tau$ and to $-1$ for 
$t>\tau$ for some $\tau\in\,]a_*,b_*[$. 
\end{itemize}

\medbreak\noindent For a problem where the control set $U$ is a
compact convex subset of $\rr^m$, a control $\eta_*$ that takes values
in the set of extreme points of $U$ is said to be a {\bem bang-bang
control}.  In our case, we have proved that {\bem all optimal
controls are bang-bang and either constant of piecewise constant 
with one switching}. 

Notice that the switchings of the optimal control $\eta_*$ are determined
by the function $\varphi$, which in this example happens to be 
$\pi_y$. That is why this function is called the {\bem switching function}
for this problem.

Notice also that {\bem we have made essential use of all the conditions
given by Theorem \ref{theo-timevar}}. In particular, 
{\bem the nontriviality condition was crucial, and the fact 
that the value of the Hamiltonian is equal to zero was also decisive.}


\bibliographystyle{amsalpha}


}

\end{document}

